Author’s response to reviews

Title: Brevibacterium paucivorans bacteremia: Case report and review of the literature

Authors:

Nobuhiro Asai (nobuhiro0204@hotmail.com)
Hiroyuki Suematsu (hsuemat@aichi-med-u.ac.jp)
Atsuko Yamada (atsuy@aichi-med-u.ac.jp)
Hiroki Watanabe (watanabe.hiroki.769@mail.aichi-med-u.ac.jp)
Naoya Nishiyama (n.naoyaso@gmail.com)
Daisuke Sakanashi (saka74d@aichi-med-u.ac.jp)
Hideo Kato (katou.hideo.233@mail.aichi-med-u.ac.jp)
Arufumi Shiota (shiota@aichi-med-u.ac.jp)
Mao Hagihara (hagimao@aichi-med-u.ac.jp)
Yusuke Koizumi (ykoizumi@aichi-med-u.ac.jp)
Yuka Yamagishi (y.yamagishi@mac.com)
Hiroshige Mikamo (mikamo@aichi-med-u.ac.jp)

Version: 4 Date: 02 Feb 2019

Author’s response to reviews:

Dear Cecilia Devoto

BMC Infectious Diseases

https://bmcinfectdis.biomedcentral.com/

Thank you for your reviewing my article. We made some changes according to your suggestions. The portions revised were red. It seems same as revision 1. We strongly believed that this
manuscript became better than before. Please, re-consider this article for publication in your journal.

Sincerely,

Nobuhiro Asai, Hiroshige Mikamo  31 Jan 2019